

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

An Agent-Based Model of Tsetse Fly Response to Seasonal Climatic Drivers: Assessing the Impact on Sleeping Sickness Transmission Rates

Citation for published version:

Alderton, S, MacLeod, E, Anderson, N, Palmer, G, Machila, N, Simuunza, M, Welburn, S & Atkinson, PM 2018, 'An Agent-Based Model of Tsetse Fly Response to Seasonal Climatic Drivers: Assessing the Impact on Sleeping Sickness Transmission Rates' PLoS Neglected Tropical Diseases, vol. 12, no. 2, e0006188. DOI: 10.1371/journal.pntd.0006188

Digital Object Identifier (DOI):

10.1371/journal.pntd.0006188

Link:

Link to publication record in Edinburgh Research Explorer

Document Version: Publisher's PDF, also known as Version of record

Published In: PLoS Neglected Tropical Diseases

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





Citation: Alderton S, Macleod ET, Anderson NE, Palmer G, Machila N, Simuunza M, et al. (2018) An agent-based model of tsetse fly response to seasonal climatic drivers: Assessing the impact on sleeping sickness transmission rates. PLoS Negl Trop Dis 12(2): e0006188. https://doi.org/10.1371/ journal.pntd.0006188

Editor: Philippe Solano, Institut de recherche pour le developpement, FRANCE

Received: August 17, 2017

Accepted: December 22, 2017

Published: February 9, 2018

Copyright: © 2018 Alderton et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: SA is supported by an EPSRC Doctoral Training Centre grant (EP/G03690X/1). PMA and SCW are supported by, and the fieldwork in this investigation was carried out for, the Dynamic Drivers of Disease in Africa Consortium, NERC project no. NE/J000701/1, part of the Ecosystem Services for Poverty Alleviation (ESPA) RESEARCH ARTICLE

An agent-based model of tsetse fly response to seasonal climatic drivers: Assessing the impact on sleeping sickness transmission rates

Simon Alderton^{1,2,3}*, Ewan T. Macleod⁴, Neil E. Anderson⁵, Gwen Palmer⁶, Noreen Machila^{4,7}, Martin Simuunza⁷, Susan C. Welburn⁴, Peter M. Atkinson^{1,2,3,8,9}

 Geography and Environment, Faculty of Social and Human Sciences, University of Southampton, Southampton, United Kingdom, 2 Lancaster Environment Centre, Lancaster University, Lancaster, United Kingdom, 3 Centre for Health Informatics, Computing and Statistics (CHICAS), Lancaster Medical School, Lancaster University, Lancaster, United Kingdom, 4 Division of Infection and Pathway Medicine, Biomedical Sciences, Edinburgh Medical School, College of Medicine and Veterinary Medicine, The University of Edinburgh, Edinburgh, United Kingdom, 5 The Royal (Dick) School of Veterinary Studies and the Roslin Institute, University of Edinburgh, Roslin, United Kingdom, 6 Independent Researcher, Leyland, Lancashire, United Kingdom, 7 Department of Disease Control, School of Veterinary Medicine, University of Zambia, Lusaka, Zambia, 8 School of Geography, Archaeology and Palaeoecology, Queen's University Belfast, Northern Ireland, United Kingdom, 9 State Key Laboratory of Resources and Environmental Information System, Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing, China

* s.alderton@lancaster.ac.uk

Abstract

Background

This paper presents the development of an agent-based model (ABM) to incorporate climatic drivers which affect tsetse fly (*G. m. morsitans*) population dynamics, and ultimately disease transmission. The model was used to gain a greater understanding of how tsetse populations fluctuate seasonally, and investigate any response observed in *Trypanosoma brucei rhode-siense* human African trypanosomiasis (rHAT) disease transmission, with a view to gaining a greater understanding of disease dynamics. Such an understanding is essential for the development of appropriate, well-targeted mitigation strategies in the future.

Methods

The ABM was developed to model rHAT incidence at a fine spatial scale along a 75 km transect in the Luangwa Valley, Zambia. The model incorporates climatic factors that affect pupal mortality, pupal development, birth rate, and death rate. In combination with fine scale demographic data such as ethnicity, age and gender for the human population in the region, as well as an animal census and a sample of daily routines, we create a detailed, plausible simulation model to explore tsetse population and disease transmission dynamics.

Results

The seasonally-driven model suggests that the number of infections reported annually in the simulation is likely to be a reasonable representation of reality, taking into account the high



programme. The ESPA programme is funded by the Department for International Development (DFID), the Economic and Social Research Council (ESRC) and the Natural Environment Research Council (NERC). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

levels of under-detection observed. Similar infection rates were observed in human (0.355 per 1000 person-years (SE = 0.013)), and cattle (0.281 per 1000 cattle-years (SE = 0.025)) populations, likely due to the sparsity of cattle close to the tsetse interface. The model suggests that immigrant tribes and school children are at greatest risk of infection, a result that derives from the bottom-up nature of the ABM and conditioning on multiple constraints. This result could not be inferred using alternative population-level modelling approaches.

Conclusions

In producing a model which models the tsetse population at a very fine resolution, we were able to analyse and evaluate specific elements of the output, such as pupal development and the progression of the teneral population, allowing the development of our understanding of the tsetse population as a whole. This is an important step in the production of a more accurate transmission model for rHAT which can, in turn, help us to gain a greater understanding of the transmission system as a whole.

Author summary

African trypanosomiasis is a parasitic disease which affects humans and other animals in 36 sub-Saharan African countries. The disease is transmitted by the tsetse fly, and the human form of the diseases is known as sleeping sickness. In an attempt to improve our understanding of the mechanisms which contribute to sleeping sickness transmission, a detailed, seasonally driven model of the tsetse fly has been produced, with the theory that a greater understanding of the disease vector's life cycle will allow developments in our knowledge of disease transmission. The model incorporates previously developed spatial data for the Luangwa Valley case study, along with demographic data for its inhabitants. Tsetse and potential human and animal hosts are modelled at the individual level, allowing each contact and infection to be recorded through time. Through modelling at a fine-scale, we can incorporate detailed mechanisms for tsetse birth, feeding, reproduction and death, while considering what demographics, and which locations, have a heightened risk of disease.

Introduction

The tsetse fly (genus: *Glossina*) is the vector for human African trypanosomiasis (HAT) or sleeping sickness, a neglected tropical disease caused by two sub-species of the protozoan parasite *Trypanosoma brucei* s.l.: *T. b. rhodesiense*, in eastern and southern Africa and *T. b. gambiense* in West Africa [1]. *T. b. rhodesiense* HAT (rHAT) is a zoonosis, affecting a wide range of wildlife [2,3] and domestic animals, particularly cattle [4], presenting in humans as an acute disease [5]. The history of HAT in sub-Saharan Africa is characterised by long periods of endemicity where the disease self-sustains at low background levels, with periodic epidemics in regional *foci* [6]. As sleeping sickness is a neglected tropical disease, treatments are often out-of-date, difficult to administer, physically invasive and partially validated, with the prospect for future developments of more effective treatments being limited (e.g. [7–11]). Furthermore, where tools are available, HAT is rarely prioritised due to competing public health interests [12]. In terms of disease prevention, there is currently no immunological prophylaxis to stop infection in humans [13], made difficult to produce due to the parasite being able to

evade the host's immune response by altering the antigenic character of its glycoprotein surface coat [14]. Given these difficulties with preventing and treating HAT infection in humans, it is not surprising that mitigation strategies focused on vector control have seen success (e.g. [15–18]), given that the tsetse fly is not only required for transmission, but also for several stages of parasite development [19,20]. Despite such efficacy, the control of the disease in tsetse (and, therefore, wildlife) in game reserves and other protected areas is complicated by ecological, conservationist and environmental considerations [21–23].

Gaining a greater understanding of the population dynamics in a tsetse population appears to be an attractive goal, considering that such an understanding could lead to the development of more targeted vector control strategies which have a less adverse ecological impact, while also allowing a more plausible understanding of the rHAT transmission system. For the latter, demographic growth (through the availability of food and habitat) and climate changes (affecting tsetse development and mortality rates) are two factors which could affect tsetse population dynamics, and ultimately affect the transmission system [24,25]. As a result of the significant role that a tsetse population has in determining the rate and distribution of rHAT transmission, this paper considers the tsetse sub-component of the larger rHAT transmission system in detail, with the ultimate goal being the creation of a more accurate representation of the transmission system as a whole.

Collecting comprehensive data on populations of tsetse in the field is expensive, complex and time consuming and, consequently, numerous attempts have been made to model tsetse populations as part of vector control or HAT transmission studies (e.g. [26-29]). Some models incorporate climatic drivers which create fluctuations in the tsetse population through the seasons (e.g. [30-33]). One recent example used agent-based modelling (ABM) techniques to simulate a simple fluctuation in tsetse population size through different seasons by altering the length of a predetermined lifespan for tsetse, depending on whether the tsetse emerges in the dry (2 months) or wet season (3 months) [31]. Incorporating more detail, [33] used known relationships between temperature and different life events and processes, such as mortality and the length of the pupation period, as parameters when constructing a population model for vector control.

ABMs are "a computerized simulation of a number of decision-makers or agents, and institutions, which interact through prescribed rules" [34]. ABMs have been described as a "third way" of conducting scientific research, incorporating both deductive since ABMs start with basic assumptions, and inductive approaches, as they produce simulation data to analyse [35]. However, Epstein [36] suggests that rather than inductive or deductive, ABMs should be considered as "generative" tools in that, through the initialisation of a population of autonomous agents in a relevant spatial environment, one can allow the agents to interact given a simple set of local rules, and generate, from the bottom up, the macroscopic behaviour and regularity of the population as a whole. Such an approach lends itself well to both the investigation of the HAT transmission system as a whole and the tsetse populations and their dynamics as a component. Starting with tsetse population dynamics, much is written about how varying climatic conditions have different impacts on various tsetse life events and processes e.g.: pupal period duration (e.g. [37]), probability of pupal death (e.g. [38,39]), and time between oviposition (e.g. [40,41]). Representing observations made from samples acquired both in the field and laboratory studies, these patterns provide us with a solid framework to model the larger population, for which comprehensive data are much more difficult, if not impossible, to acquire. By initialising a tsetse population as individuals, each abiding by rules set by the above behavioural patterns (and others relating to feeding, mating and age-dependent mortality), plausible population level outcomes such as fluctuations in population size should be observable as the simulation progresses.

When the HAT transmission system is incorporated into an ABM for acquiring preliminary knowledge of the disease transmission system, the constructed model becomes a representation of a complex system (e.g. [42–44]), given that the prevalence of the disease is a complicated emergent phenomenon produced by relatively simple, individual specific rules (both vector and host) concerning movement and resource acquisition. In a complex system, the causes of emergent phenomena cannot easily be decoupled and explained by specific parts of the system [45] with, in this case, the model landscape and agent behaviour creating variation in the timing, location and probability of infection as a result of their influence on variability in contact patterns between vector and host [46,47]. In this way, ABMs could be considered the most appropriate way to investigate both the HAT transmission system, and tsetse fly dynamics as a subcomponent, allowing the representation of interdependent processes such as how individuals interact with each other and their environment through space and time more easily than is possible through more traditional epidemiological techniques [48].

In previous work, an ABM of rHAT transmission was produced using a spatialized approach, incorporating factors often overlooked (e.g. human behaviour and activity-based movement; density and mobility of vectors; and the contribution of additional hosts) [27]. This paper presents the first ABM which considers the effect of climatic factors on individual tsetse and their life processes in detail, while also considering the effect this has on rHAT transmission in a large study area in Eastern Province, Zambia. Through the incorporation of seasonality parameters into an existing fine spatial and temporal scale ABM of rHAT transmission in the region [27], the aim was to develop a greater understanding of tsetse population dynamics through simulation, and subsequently produce a more plausible model of rHAT transmission. The incorporation of such data is vital where transmission rates, and indeed the transmission system as a whole, are to be explored over multiple years. The existing model provided a suitable starting point for the simulation of these seasonal parameters by modelling tsetse flies at the individual level, along with different life events for which durations and probability of occurrence can be climatically constrained. Ultimately, the modified model was implemented with the aim of answering the following research questions: throughout the year, how does the tsetse fly population fluctuate both as a whole, and within different life stages (e.g. pupal, teneral, mature)? Under the caveat that a plausible model has been produced, what rates of disease transmission are observed, and how do these vary seasonally? Such a model will allow for future exploration of long-term mitigations strategies, alterations to the demographic make-up of the study area, and climate change scenarios.

Methods

Study area

Eastern Province, Zambia is situated in southern Africa, sharing borders with Malawi (to the East) and Mozambique (to the South). The Luangwa Valley is an extension of the Great Rift Valley of East Africa, traversing the Zambian Eastern, Northern and Muchinga Provinces. The valley is characterised as a flat bottomed valley bounded by steep, dissected escarpments which rise to a plateau at approximately 900–1000 m [49]. Different types of vegetation are observed at different altitudes, with valley areas consisting mainly of mopane woodland and patches of grassland, while the natural vegetation on the escarpment and plateau is miombo woodland, interspersed with munga woodland [50].

The study area spans a sparsely populated region of the Luangwa Valley. Villages are small (between 5 and 20 households) and inhabitants are predominantly subsistence farmers. The data collection area and region to be modelled consists of a 75 km transect which starts close to Mfuwe airport in the north, and runs southwards along the Lupande River and its

distributaries (Fig 1). Average monthly temperature and rainfall measurements collected at the Mfuwe airport (1982–2012) weather station are reproduced in Fig 2 [51]. There are three main seasons in Zambia's tropical climate: the rainy season spans November to April (wet and



Fig 1. Map of the study area size and location. Households from the census included in this modelling study are indicated as white circles and Mfuwe airport, the location of the region's weather station, is visible in the north (produced using Landsat 7 imagery from USGS), after [27].

https://doi.org/10.1371/journal.pntd.0006188.g001





warm) with mean monthly rainfall peaking at 210 mm in January. After the rains, a cold and dry period occurs prior to August, in which May is the hottest and wettest month, with mean temperatures below 23°C and mean rainfall below 3 mm. The hot and dry season usually spans August, September and October, with mean temperatures reaching 28°C in October accompanied by 17 mm of rainfall on average, the first after four dry months in succession [49,51]. The Luangwa River and its main tributaries are perennial, and although flash flooding occurs in all rivers during the wet season, the smaller rivers which drain the valley floor dry out during the dry season and flow during the rains [52].

rHAT is endemic in the Luangwa Valley, first being reported in 1908 [53]. *G. m. morsitans* was not originally considered a vector of rHAT in the valley, despite 50% of domestic and game animals in the Valley having been observed to harbour trypanosomes [54]. In the early 1970s, a large rHAT outbreak occurred in Isoka (241 case in 3 years) attributed to fly encroachment from Luangwa [55]. Wildlife had been observed to reside in Isoka for several months during the rainy season, migrating away during the dry season. In 1973, early diagnosis and improved treatment methods were introduced, and case numbers fell [56].

Today, cases of rHAT continue to be reported in the Luangwa Valley. Mid-Luangwa Valley has recently experienced increased immigration of people seeking fertile land. Land pressure has resulted in human settlement in increasingly marginal, tsetse-infested areas, previously avoided for fear of disease risk to introduced livestock. Households grow cotton as a cash crop and maize and groundnuts for home consumption [49]. These anthropogenic changes have the potential to destabilise current trypanosomiasis transmission cycles, resulting in increasing prevalence of trypanosomiasis in both human and animal hosts, and the spread of rHAT into previously unaffected areas. Risk factors include human proximity to the large wildlife reservoir in the South Luangwa National Park to the north-west [2], and ever-increasing livestock and human density on the plateau. Little is known concerning tsetse-trypanosome-human interaction in the region. Therefore, the ABM has the potential to enable exploration of contact risk within communities. Furthermore, with climate changes expected to occur in the near future, such as reduced annual rainfall, increased storm events and increased temperature [57][58], it is becoming increasingly important to understand how climate factors can affect tsetse populations, particularly in areas such as this, where increases in temperature could see the tsetse habitat spreading further up the valley to more populous areas.

Adding seasonality-The exploration of climatic drivers

This paper describes a new, seasonally sensitive ABM for rHAT/animal African trypanosomiasis (AAT), based on an earlier, non-seasonal model that was constructed using data derived from a detailed rHAT, AAT, and *G. m. morsitans* ecological survey, undertaken in 2013, in Eastern Province, Zambia [27]. Due to the fine spatial and temporal scales used to model the system, and the number of mechanisms incorporated (e.g., tsetse reproduction, tsetse feeding, human agent movements using real-world routines and pathfinding techniques [59]), the model was complex and its data inputs were numerous. As a result, only new data and modifications to the original model are described here. A detailed description of the original, nonseasonal model framework, and the data used to construct it, can be found in [27].

Temperature data. Annual mean temperature data collected at Mfuwe airport are presented in Fig 2. Fig 3 shows the interpolation of these data into daily temperatures using a four-term, sum of sines method, produced using Matlab's curve-fitting tool. This method produced the best fit to the mean monthly data ($r^2 = 0.97$), while also producing limits which join when wrapped, allowing multiple years to be considered. Future mention of temperature in this section refers to the daily temperatures derived from this curve. The temperature model is presented starting in August as this reflects the month in which the present simulation commences, chosen as the midpoint between the tsetse surveys used to estimate the tsetse population (June and November–see [27]).

Seasonal variation in oviposition. The non-seasonal model used the expectation that a female tsetse would give birth 18 days after mating, and every 10 days thereafter for the duration of their life [60]. However, data collected in Zimbabwe in 1994 suggest that in warmer conditions (~30°C), the time taken to produce a first pupa can be as low as 15 days [40], with extrapolations of the data suggesting that subsequent offspring could be produced at an interval of 16 days at 16°C, and 7 days at 31°C [41] (see Fig 4) using Eq 1 [41]:

Birth Interval =
$$\frac{1}{k_1 + k_2(T - 24)}$$
, Eq 1

Where: T = temperature, k_1 = 0.061 and k_2 = 0.0020 for the initial birth interval, and k_1 = 0.1046 and k_2 = 0.0052 for further births.



Although extrapolation should be treated with caution, the tsetse species under investigation, along with the vegetation, are consistent in both studies. Furthermore, the temperatures in the region of the present study are largely in the range of temperatures for which field experimental data were collected to produce these curves ($22^{\circ}C$ to $30^{\circ}C$) and, as a result, are used to dictate birthing intervals in the current simulation.

Temperature and pupal duration

The previous iteration of the model included a longer pupal duration in males than in females, as suggested in the literature (e.g. [37,60]), and so for each larva deposited during the simulation, a 35 and 30 day pupal period was included for males and females, respectively, represented as a period of inactivity. However, pupation is known to be temperature sensitive with pupal periods decreasing with increasing temperature, a relationship observed by Phelps and Burrow's laboratory experiments at constant temperatures [37]. Hargrove [41] utilised the data to present a near perfect fit for pupal duration at temperatures between 16°C and 32°C



Fig 4. *Tsetse reproduction intervals in response to temperature (produced using data from* [41]). https://doi.org/10.1371/journal.pntd.0006188.g004

 $(r^2 = 0.998)$ (see Fig 5), represented by Eq 2:

$$\mathsf{r} = \frac{k_3}{1 + e^{(a+bt)}},$$

$$pupal duration = \frac{1}{r}, Eq 2$$

Where: *t* = temperature, for males: a = 5.3, b = -0.24 and $k_3 = 0.053$ and for females: a = 5.5, b = -0.25 and $k_3 = 0.057$.

Given the excellent fit to the data and the large variation in pupal periods expected within the temperature range found in the study area ($19^{\circ}C = -60$ days, $28^{\circ}C = -20$ days), variation in pupal duration with temperature is clearly an important factor to incorporate in the model.

Climatic drivers for pupal mortality. Due to the difficulty in finding pupae in the field, population estimates, and ultimately estimates of pupal death rates are rare [41]. Although observations suggest temperature may not affect pupal mortality until extremes are reached (>35°C) [39], pupal mortality in tsetse flies is shown to vary seasonally [61]. Similar to Vale



Fig 5. Tsetse pupal intervals in response to temperature (produced using data from [41]).

NEGLECTED TROPICAL DISEASES

> and Torr's [29] climate independent tsetse model, the previous iteration of this ABM assumed a pupal mortality rate of 26%, after field observations by Jackson [62]. However, as a result of the implementation of a variable pupal duration in the current version, it seems appropriate to apply pupal mortality rate on a daily basis. While the 1% daily pupal mortality rate field estimates presented by Rogers and Randolf [63] were used as a starting point here, Ackley and Hargrove [61] found that mortality rates in immature fly stages peak before and after the rainy season, with model output for G. pallidipes reporting a peak mortality rate of 0.12 during March in Rekomitjie, Zimbabwe. To capture this apparent variability, the pattern presented was adapted and applied to the current investigation. Modifications were required as the pattern of mortality presented by Ackley and Hargrove considers pupae and immature flies together, while the model in this study models pupae separately-resulting in a risk of double counting mortality in immature flies. In addition, the species of tsetse fly under investigation here is G. m. morsitans rather than G. pallidipes—G. pallidipes pupae are found to show greater susceptibility to extreme temperatures (e.g. [64]). As a result of these factors, simply applying the same mortality rates to the current investigation would overestimate pupal mortality, with tests showing that the values of daily mortality reported by Ackley and Hargrove needed to be



Fig 6. The upper two plots present an observed relationship between tsetse mortality rate (MR) and temperature (produced using data from [41]). The lower two plots illustrate a modelled relationship between tsetse age and mortality rate (produced using data from [65]). Blue = male tsetse. Red = female tsetse.

NEGLECTED TROPICAL DISEASES

scaled to 60% to provide stability in the current investigation. Mortality rates were applied such that each day, each pupa had a probability of dying based on these scaled mortality rates.

Combining age and temperature dependent mortalities. The first iteration of the model included a tsetse death function to deal with mortality not captured by starvation of individual tsetse, or pupal mortality. Represented as a scaled-down version of Hargrove's age-dependent mortality model [65], no climatic impact was incorporated. To ensure the tsetse agents are sensitive to temperature in the current iteration of the model, the previous age-dependent mortality rate (Eq 3) [65] was adapted to reflect the relationship reported by Hargrove in Eq 4 [41], a modelled estimate of daily mortality rate in response to temperature based on mark-recapture data collected on Antelope Island in Zimbabwe in 1980 and 1981. Eqs 3 and 4 are visualised for both male and female tsetse in Fig 6.

$$\lambda_1(t) = k_4(k_5 e^{-k_5 a(t)} + k_6 e^{k_6 a(t)}),$$
 Eq 3

Where: *t* is a point in time, a(t) is fly age at time *t*, and k_4 , k_5 , k_6 are gender-specific constants (Male: $k_4 = 0.389$, $k_5 = 0.395$, $k_6 = 0.0583$ and female: $k_4 = 0.605$, $k_5 = 0.201$, $k_6 = 0.0119$) [65].

$$\lambda_2(t) = rac{e^{-k_7 + (k_8 T(t))}}{100.0},$$
 Eq.4

Where: T(t) is the temperature at time *t*, and k_7 , k_8 are gender-specific constants (Male: $k_7 = 0.19$, $k_8 = 0.071$ and female: $k_7 = 0.85$, $k_8 = 0.083$) [41].

Both Eqs 3 and 4 are hazard functions, which define the likelihood that something will survive to a certain point in time based on its survival at an earlier time, each taking a different factor into account. Examples of combining hazard functions are well recorded in reliability engineering, where "failure" is used in place of "hazard" or "mortality", and the failure rate of a series system is calculated as the sum of the failure rates of its components [66]. While more complicated methods are required to calculate failure rates of parallel and combined systems, in complex systems, where relationships between two components cannot be defined easily, the system's failure rate is pessimistically taken to be the sum of the individual failure rates of its components. Since the relationship between age and temperature on the tsetse mortality rates are not well defined, the latter method has been implemented for this study (Eq 5).

$$A(t) = \sum_{i=1}^{n} \lambda_i(t), \qquad \text{Eq 5}$$

where, $\lambda_i(t)$ is the hazard rate for 'component' *i*.

While the temperature-dependent mortality rate $(\lambda_2(t))$ will have been determined across a range of fly ages, the age-dependent mortality rate $(\lambda_1(t))$ was determined during a relatively small window in which the temperature was likely to be constant. To avoid 'double counting' the mortality rate due to temperature, the age-dependent mortality rate was decoupled from temperature to remove any associated effects, via the new function in Eq.6. While this is relatively crude, this was used as a starting point in the absence of a more detailed understanding of the relationship between temperature and age in the mortality of the tsetse fly.

$$\lambda_1(t) = k_4 \left(k_5 e^{-k_5 a(t)} + k_6 e^{k_6 a(t)} \right) - \frac{e^{-k_7 + (k_8 T_c)}}{100.0},$$
 Eq 6

where, T_c is a constant temperature, which could be considered as the 'base' temperature.

Using this equation and the constant values reported previously, the daily mortality rate of a tsetse fly at a given age and temperature can be visualised as in Fig 7 for males and Fig 8 for females.

ABM framework

The previous, non-seasonal ABM provided the majority of the methods and data used in the current version of the simulation, and so readers are referred to [27] for greater detail and only a summary is provided here. Census data were used to locate and initialise the human and animal populations living in the households shown in Fig 1. A sample of resource-seeking routines sorted by gender and age was taken in the field (see supplementary information of [27]), and a set of plausible paths from each village to each resource was created using a pre-processing A* pathfinding technique [59]. For tsetse, an estimate of the total apparent population size, density and distribution was provided. Four agent types were included in the ABM, together with an areal representation of wildlife. Humans, cattle, other domestic animals and tsetse used in the ABM were constructed as four separate classes, with populations modelled 1:1 with the data collected in the census (e.g. 16,024 human agents) and the estimated tsetse population discussed previously. Each class had its own initial information and storage structures for



Fig 7. Male tsetse mortality rate dependent on age and temperature.

events that occurred through the simulation. The ABM was written in Python 2.7 using an object-oriented framework, and run on the Lancaster University High End Computing (HEC) Cluster, with all spatial data being processed using Quantum GIS 1.8.0.

The subsequent sections draw attention to any modifications between the original, non-seasonal modelling framework and the new ABM model, while also describing how the climatic drivers affecting the tsetse population were incorporated into the model.

Temporal resolution and wildlife feeds

The initial iteration of the model was split into 2,400 time-step (or tick) days, as the more frequent the tick, the smaller the jumps made by agents as the simulation updates, and the less chance of missing potential interactions. However, this method was restrictive in terms of memory usage and CPU time required to run just six months of the simulation. Further tests were carried out to establish how coarse the temporal resolution could be made before the number of simulated domestic host-vector contacts was reduced, and a greater daily probability of wildlife feed was required to maintain the tsetse population levels. It was established that 600 ticks per day (2.4 minutes per tick) allowed the simulation to progress with no obvious effect on human, cattle and other domestic animal bite numbers, while requiring a very similar daily wildlife bite probability to produce a stable tsetse population (37% chance per day of a hungry tsetse taking a wildlife bite, compared with 35% in the previous version). As a result, 600 ticks per day were used to produce the results of this investigation, which required approximately 4.5 GB of RAM per simulation run on the high performance machine, and 24 hours of CPU time per simulated year.



Fig 8. Female tsetse mortality rate dependent on age and temperature.

Climatic driver additions

To capture the effect of seasonality on the tsetse fly population, daily temperature was calculated every 24 hours using the interpolation method discussed previously, and set as a global variable for the simulation.

Tsetse births

For each female, once mated, the number of days since mating was compared with the birth interval calculated using Eq 1 and the daily temperature. If and when the number of days since mating exceeded the interval calculated on a given day, a pupa was deposited. A count of the number of days since mating was replaced with a count of the number of days since last off-spring, and Eq 1 was used again on a daily basis (using the alternative constants for further births), until another birth occurred. This process was repeated for the duration of a female tsetse fly's lifespan. There was an equal chance of each tsetse offspring being male or female, and each pupa was deposited in a bush area in which the female tsetse rested during the previous night.

Pupal duration and death

A rolling average of the temperature that each pupa has experienced since birth was calculated and attributed to each individual. This temperature was used to determine each individual's pupal duration, given that if a pupa's age exceeded the pupal duration calculated using Eq 2, the pupa would emerge as a teneral fly. It was considered important to use a rolling average of

temperature here as the length of a pupal period can span months with quite different temperatures. As described previously, rather than a single probability used to decide whether a pupa would die during its entire pupal period, a variable daily probability of pupal death was included, increased in some months to account for losses observed in the rainy season. Should the probability be exceeded for a pupa, that tsetse was removed from the simulation.

Teneral and mature tsetse death

Death could result from pupal mortality, starvation, or if a tsetse fly exceeded the daily mortality rate calculated by sex, age and temperature (Eq 6, Figs 7 and 8). The mortality rate was calculated individually for each teneral and mature fly, and if the probability was exceeded, the tsetse was removed from the simulation. Starvation occurred if a tsetse tried and failed to feed before a given period of time had elapsed. The starvation element was more strict for teneral flies (3 days instead of 5 days) highlighting their increased vulnerability and reduced flight strength. In the previous version of the simulation, 75 teneral files were added to the simulation for the first 35 days to account for pupae deposited prior to the start of the simulation. As this version of the simulation started in August, and the simulated climate quickly became hostile for teneral flies as temperature increased, 500 teneral tsetse were required per day for the first 45 days, which is representative of average simulated pupal maturation rates during September as the simulation progressed (see Results). In the original model, in the absence of climatic factors, a scaling factor for adult fly mortality was required to offset fly starvation within the simulation. This value was set at 55%. Although this scaling factor is still required in this iteration of the model due to the same starvation element, the incorporation of temperature dependent mortality, and more detailed mechanisms for modelling pupae, has reduced the required level of scaling to 80%

To allow the model to initialise and stabilise, the simulation was run for a year before the results for this paper were produced, allowing a 'burn-in' period. For example, the results presented below are representative of years 2–4 of the simulation. 100 repeat simulations were used to produce the results presented here.

Results

Tsetse population

At the end of the three year simulation, a relatively stable population record was observed in both the male and female tsetse populations, with both exhibiting a double peak in response to the climatic driver (Fig 9).

Each year, until peak temperature was reached in October and November, the population slowly increased, with each gender's population size increasing by approximately 2000 flies. Such population increases during this hot and dry season could be attributable to the absence of a boosted pupal mortality which is observable during the rainy season [61], with increasing temperatures having a greater impact in reducing pupal duration and the period between births, than increasing tsetse teneral and mature tsetse mortality. During the rainy season (November-April), this population gradually fell to an annual low, a result of peaks in pupal mortality at the start of the rainy season, and high temperatures causing increased mortality in the annual peak population of teneral flies (see Fig 10) (now emerged after a high period of births discussed previously—birth numbers can be seen in Fig 11).

During this period, with a reduced number of pupae to develop, and teneral tsetse to mature and start reproducing, the higher temperatures no longer aided a growth in population as there were fewer pupal maturations and birth rates to 'accelerate' (Fig 11) At the end of the rainy season, the tsetse population gained a small boost due to a plateau in temperature, and



Fig 9. Average daily population size of adult tsetse flies by sex (+/- standard deviation), overlain on mean monthly temperature and precipitation for the three year simulation period.

NEGLECTED TROPICAL DISEASES

PLOS

the drop in population slowed through the cool and dry season (May to July), although recovery did not start during this period as temperatures were too low to aid rapid repopulation of the tsetse, and the pupal population was still recovering (Fig 10).

Fig 12 presents the different possible modes of tsetse death included in the model, and how the rates varied as the simulation progressed. Non-starvation death represented the deaths attributable to the age-temperature dependent mortality model defined by Eq 6, and was consistently responsible for the largest number of daily deaths, peaking in the period of highest temperature with approximately 350 deaths per day. Unsurprisingly, given its temperature dependency, the mortality shape closely aligned to mean monthly temperature, except for a period in February and March after the pupal population was reduced by a period of high pupal mortality during the rainy season, resulting in a reduced teneral population and, therefore, fewer adult deaths. Deaths due to starvation followed closely the general pattern of population size, with teneral starvations being particularly low–likely a result of the low daily teneral population size (ranges between 100 and 400 –Fig 10) and the teneral tsetse population having the highest age-temperature dependent mortality rate. Using the Ackley and Hargrove model [61] for pupal mortality produced peaks prior to the rainy season and, to a lesser extent, after the wettest months (Fig 12).

PLOS



Fig 10. Average daily population size of tsetse flies at different life stages, overlain on mean monthly temperature and precipitation for the three year simulation period.

https://doi.org/10.1371/journal.pntd.0006188.g010

The ratio of pupae to mature tsetse was approximately 2:1 at any given time, with the mature to teneral population ranging between 15:1 at the peak of population size and 25:1 when population sizes were generally lower.





NEGLECTED TROPICAL DISEASES

PLOS

rHAT transmission

Across the three year simulation, the approximate incidence rate for human and cattle rHAT infections was 0.355 per 1000 person-years (SE = 0.013), and 0.281 per 1000 cattle-years



Fig 12. Average daily deaths by class for the tsetse population, including starvation of both teneral and mature flies, pupal mortality, and other deaths attributable to the age and temperature dependent mortality model discussed previously. Error bars represent the standard deviation.

(SE = 0.025). There were 11 human infections each year on average (i.e. per year, per run), and 2 cattle infections.

Fig 13 illustrates how these infections clustered spatially and by season. The aggregate number of infections across all years and each of the 100 repeats was used to produce this heat map



Fig 13. Heat surface representing aggregate number of infections across the 100 repeat simulations, by season, with pixel values taking the units of infections per square kilometre.

due to the low infection numbers. There was not much spatial variability through the seasons despite the variation in tsetse population size.

State	Total (3 years)		Hot, Dry Season		Hot, Wet Season		Cool, Dry Season	
	Average	SE	Average	SE	Average	SE	Average	SE
School	50.0%	1.2%	49.5%	2.1%	50.1%	1.7%	51.1%	2.8%
Resting	3.2%	0.3%	3.7%	0.8%	3.4%	0.5%	2.4%	0.6%
Market	0.1%	0.1%	0.4%	0.2%	0.1%	0.1%	0.0%	0.0%
Firewood	5.8%	0.5%	5.9%	1.0%	6.0%	0.7%	3.8%	0.7%
Farm	15.3%	0.8%	14.8%	1.5%	16.0%	1.1%	16.9%	2.2%
Water	25.5%	1.2%	25.7%	1.9%	24.4%	1.6%	25.8%	2.3%
Graze Cattle	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%
Water Cattle	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%

Table 1. Average proportion of human infections attributable to each activity, for each season, represented as a percentage.

https://doi.org/10.1371/journal.pntd.0006188.t001

However, the number of infections reduced during the second half of the rainy season with the lowest density of infections observed during the cool and dry months. Two hotspots are visible in each of the seasons, each with elongated elements suggesting that frequently used paths were sources of interaction between vector and human host. This is possibly most visible in the north as east-to-west movement here could represent movement between villages and the river, a hypothesis which is given support by observations of infection by activity (Table 1) which suggest that in each season, water collection accounted for approximately 25% of human infections, second only to school trips which accounted for 49% to 51% of infections. No human infections were acquired whilst watering or grazing cattle, while the third highest number of infections occurred when farming. There was little variation in infections by activity ity between the seasons.

With the observed high proportion of infections coming from school trips, it is unsurprising that 5–10 year olds and 10–18 year olds had the highest infections rates (Table 2). Infection rates were generally lower in the cool and dry season, peaking in the hot and dry season.

Table 3 shows that the highest incidence rates were observed amongst immigrant tribes, with the only indigenous tribe (the Kunda) exhibiting one of the lowest infection rate across each time period, despite making up over 70% of the population.

Infection rates observed by gender and cattle ownership were comparable across time periods, with males and cattle owning households exhibiting marginally higher infections rates in comparison to females and households without cattle (Table 4).

Infections acquired and matured within the tsetse population fluctuated as the three year simulation progressed, with a small year-on-year increase in average infections both in the midgut and salivary gland (Fig 14). On average, the peak time of salivary gland infection development was at the beginning of the rainy season, which reflects the period of highest tsetse densities plus a time-lag for development of mature infections in the fly.

11 0001										
Age	Total (3 years)		Hot, Dry Season		Hot, Wet Season		Cool, Dry Season			
	Average	SE	Average	SE	Average	SE	Average	SE		
age1-5	0.0241	0.0037	0.0231	0.0072	0.0262	0.0058	0.0210	0.0070		
age5-10	0.7000	0.0302	0.7535	0.0484	0.7153	0.0365	0.6117	0.0504		
age10-18	0.6456	0.0277	0.7038	0.0437	0.6323	0.0323	0.6107	0.0445		
age18-60	0.2396	0.0116	0.2697	0.0223	0.2365	0.0150	0.2151	0.0168		
age60+	0.1850	0.0246	0.3169	0.0581	0.1717	0.0331	0.0792	0.0282		

Table 2. Approximate incidence rates by season for different age groups.

https://doi.org/10.1371/journal.pntd.0006188.t002

Ethnicity	Total (3 years)		Hot, Dry Season		Hot, Wet Season		Cool, Dry Season	
	Average	SE	Average	SE	Average	SE	Average	SE
Chewa	0.7678	0.0391	0.8605	0.0597	0.7503	0.0474	0.7051	0.0513
Kunda	0.2451	0.0106	0.2733	0.0176	0.2410	0.0125	0.2245	0.0151
Ngoni	2.3535	0.1939	2.4461	0.3039	2.5108	0.2432	1.8931	0.2944
Nsenga	0.0688	0.0117	0.0669	0.0214	0.0632	0.0159	0.0818	0.0234
Bemba	0.4128	0.0719	0.5654	0.1539	0.4481	0.1097	0.1888	0.1147
lenje	0.2424	0.1194	0.2424	0.2424	0.2424	0.1706	0.2424	0.2424
No Data	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000

Table 3. Approximate incidence rates by season for different ethnicities.

https://doi.org/10.1371/journal.pntd.0006188.t003

Discussion

Tsetse population

The first plausible individual-based model representation of a real world tsetse population was created allowing a simulation of the system over multiple years. The model was specified using temperature-dependent parameters derived from the literature, detailed human and animal information from acquired datasets, and expert opinion, and an estimate of the initial tsetse population size and distribution. For example, the pupal population which was completely emergent from the model (as no initial pupae data were inputted) corresponded with literature findings that pupae are comparatively difficult to find in the rainy season, and that the pupal population will be greater than that of the developed flies [38], unsurprising considering that the parameters suggest that pupae are 'safer' than teneral flies, pupal duration is at least 3 weeks, and a constant flow of developing pupae is required to replace teneral files which are dying or maturing. In addition, the ratio of female-to-male tsetse fluctuated around 2:1, a change from the simpler, non-seasonal model [27], but more in line with estimates in the literature [67], possibly as a result of running the simulation for longer, and with the addition of climate-driven parameters. The shape of the mature population was comparable to samples of tsetse collected in the region of the South Luangwa National Park (Regional Tsetse and Trypanosomiasis Control Programme (RTTCP) data reported in [22]), Eastern Province, Zambia [68], G. pallidipes in neighbouring Zimbabwe [61], and similar, yet less detailed, ABM studies [31].

The peak adult population of around 6500 flies suggests that the relatively crude technique used to extrapolate sample data from tsetse surveys for initial model construction (see [27] for more detail) produced a reasonable estimate with 5250 flies. Furthermore, the small teneral population observed is perhaps not a surprise, given that the teneral stage is a brief transition with a gradual input of developing pupae, and high mortality rates coupled with maturation to adult fly on first feed as outputs.

Gender	Total (3 years)		Hot, Dry Season		Hot, Wet Season		Cool, Dry Season	
	Average	SE	Average	SE	Average	SE	Average	SE
Male	0.3623	0.0146	0.3877	0.0222	0.3627	0.0172	0.3351	0.0240
Female	0.3486	0.0148	0.4012	0.0243	0.3453	0.0167	0.3015	0.0202
Cattle House								
TRUE	0.3979	0.0250	0.4160	0.0353	0.4150	0.0306	0.3443	0.0365
FALSE	0.3464	0.0133	0.3901	0.0205	0.3410	0.0143	0.3124	0.0189

Table 4. Approximate incidence rates by season for different sexes and cattle ownership.

https://doi.org/10.1371/journal.pntd.0006188.t004





The decrease in pupal population during the rainy season, combined with a consistently small teneral population highlights how one or two years with a very hot and wet rainy season could have serious consequences for a tsetse population, with a reduction in pupal development during periods of high mortality, and high temperatures killing more teneral tsetse reducing the birth rate over subsequent months. Similarly, such a relationship could occur over the coming years in response to climate change, with IPCC reports suggesting that more

extreme rainfall events could occur, along with a rise in temperature over the next 50 years (e.g. [57,58]). As a result, it is not surprising that some studies have suggested that certain tsetse fly populations could face extinction within the next 50 years [69]. Future studies will consider using the present model as the basis to test future climate change scenarios and examine the response in the tsetse population to such perturbations.

rHAT transmission

The model suggested similar incidence rates for rHAT infection in humans and cattle, which is likely to be a response to both the fact that the majority of the cattle were in households at the south of the transect, away from the tsetse zone (only approximately 550 of 2925 cattle were within close proximity of the tsetse zone) [27], and that humans were modelled to be much more active than cattle in the simulation, travelling more frequently away from the home. The latter point is corroborated by similar observations of human incidence rate in both cattle owning and non-cattle owning households, particularly as no human infections occurred while tending to cattle in the field or by the river. As with observations in the previous study, collecting water and school attendance provided the highest proportion of infections by some margin, and is likely to be in response to the high frequency of both trips within the simulation and, for schools, the longer distances travelled to a sparse resource, and the time of day of the trips coinciding with tsetse activity. In support of these simulated observations, a recent study of rHAT infections in Zambia found that almost half of the observed female infections were found in school-age children [70]. The data for males suggested fewer infections in children. This perhaps reflects that school attendance in the model is overestimated for the male population, and, in reality, young men may be needed to work to provide for the family at a younger age. The high incidence rates observed in immigrant tribes gives weight to the suggestion that as populations move down the plateau and into the valley, people are increasingly occupying marginal land, and increasing their exposure to the tsetse fly. As a result, future studies using the model will look to investigate how influxes of people into the region and the associated development affects the tsetse fly population in terms of habitat availability, but also how infection patterns respond to the perturbation of the system.

Six cattle infections were used to seed the model at the beginning of the simulation (along with five goats, one dog and two pigs), to reflect the estimated prevalence of T. b. rhodesiense in the sample of animals from the study transect. The model was also implemented with 10 humans infected at the simulation start, to take into account information from medical teams in the region, who suggested that there had been two reported cases in the past year, and the known high levels of under-reporting and under-detection in the area, and further afield (e.g. [22]). For example, one study suggested that levels of under-detection of rHAT could be as high as 12 cases for every one identified [71]. Furthermore, the recent study in Zambia found that, when a period of more active surveillance was adopted, the number of diagnoses increased dramatically, suggesting high levels of under-detection in the region. In addition, the investigation found that no action was taken by approximately one quarter of people showing symptoms of rHAT infection prior to diagnosis in the study, and less than half sought medical care from a health facility on first sign of symptoms [70]. As a result, given that there is no under-detection in a simulation, two cattle and 11 human infections on average per year appears plausible, especially when considering that there is currently no removal of infection from the simulation (and no reduction in activity when infected), creating a gradually increasing reservoir of infection, and an increase in tsetse infections (Fig 14).

Despite extensive effort to incorporate seasonality accurately into the simulated system, there are some omissions which were largely unavoidable here, but which should be noted.

Firstly, in reality, the spatial distribution of tsetse will change through the seasons, with tsetse concentrated in the dense woodland vegetation in the hot dry season, and more widely dispersed in the wet and cool seasons since tsetse use microhabitats to evade extremes in temperature [60,72]. Using an interpolated temperature gradient across a study area through time may allow this behaviour to be simulated, although there would be limitations as temperatures would not reflect sheltered areas utilised by tsetse. As a result, such an implementation should be used in conjunction with a variable land classification, highlighting changes in vegetation with seasons. In addition, no data were available on how human movements vary seasonally in this region at the temporal resolution being modelled, and therefore, the daily routines used are consistent through the year. Finally, it is understood that maturation rate and transmissibility of trypanosomes in tsetse varies with temperature [60], with early work in Zambia suggesting that higher trypanosome infection rates occurred in *G. morsitans* in the hot season than in the cold season [54]. However, very little research has been carried out on this subject and, within this study, transmission rates should be low enough for this to have little impact.

Bigger picture

For the first time in the field of rHAT transmission research, data produced and relationships identified in different studies, focusing on different aspects of the tsetse life-cycle and tsetse-climate interactions, have been incorporated into a single detailed ABM, creating a plausible, stable model, which can ultimately produce a reasonable estimate for transmission rates. While providing an element of validation to these individual entomological studies, through the production of tsetse population curves which closely follow those produced from data collected in the nearby South Luangwa National Park, the model represents a step towards a greater understanding of disease transmission for rHAT in this case, while also being adaptable to gHAT foci in the future. As with all models, the ABM is not without its limitations, for example, variability in tsetse feeding behaviour and preference has been incorporated, but at a basic level. However, through working towards an accurate model representation of the disease landscape one can expect to achieve a greater understanding of the rHAT transmission system, which in turn can help the devising of spatially and temporally targeted mitigation strategies in the future, to help those in need with sustainable solutions, and are more appropriate for spatially marginal communities susceptible to neglected tropical diseases.

Conclusion

The dynamics of a tsetse population are difficult to model due to difficulties in acquiring data, and the complexity of the system, but are important to understand due to their importance in rHAT transmission. Gaining a greater understanding of tsetse population dynamics may lead to greater understanding of rHAT transmission and aid future mitigation strategies. This paper presented the first seasonally-varying rHAT transmission model, defined at a fine resolution and modelling directly individual flies, with the full tsetse life cycle as a sub-component. By incorporating numerous parameters estimated from the literature, from data and from expert opinion into such a detailed model, a range of outputs were created which can be used by scientists to analyse and evaluate our current understanding of tsetse fly dynamics and the rHAT disease transmission system, and by decision-makers to investigate alternative mitigation strategies. In its current state, including seasonally varying effects, the model lends itself to modelling future scenarios, including insecticide application and other vector control strategies, the incorporation of a changing climate, the effects of landcover change and human development adjacent to, and within, the biodiverse tsetse habitat.

Acknowledgments

The authors would like to thank John Hargrove for his helpful conversations relating to tsetse fly population dynamics, and Allan Mwiinde for sharing his findings relating to the detection levels of rHAT infections in Eastern Province, Zambia.

Author Contributions

Conceptualization: Simon Alderton, Susan C. Welburn, Peter M. Atkinson.

- **Formal analysis:** Simon Alderton, Ewan T. Macleod, Neil E. Anderson, Gwen Palmer, Susan C. Welburn, Peter M. Atkinson.
- **Funding acquisition:** Noreen Machila, Martin Simuunza, Susan C. Welburn, Peter M. Atkinson.
- **Investigation:** Simon Alderton, Ewan T. Macleod, Neil E. Anderson, Noreen Machila, Martin Simuunza, Susan C. Welburn, Peter M. Atkinson.
- Methodology: Simon Alderton, Ewan T. Macleod, Neil E. Anderson, Gwen Palmer, Susan C. Welburn, Peter M. Atkinson.
- **Project administration:** Noreen Machila, Martin Simuunza, Susan C. Welburn, Peter M. Atkinson.

Software: Simon Alderton.

Supervision: Peter M. Atkinson.

Validation: Simon Alderton, Peter M. Atkinson.

Visualization: Simon Alderton, Gwen Palmer.

Writing – original draft: Simon Alderton.

Writing – review & editing: Simon Alderton, Ewan T. Macleod, Neil E. Anderson, Gwen Palmer, Noreen Machila, Martin Simuunza, Susan C. Welburn, Peter M. Atkinson.

References

- 1. Welburn SC, Fèvre EM, Coleman PG, Odiit M, Maudlin I. Sleeping sickness: a tale of two diseases. Trends Parasitol. 2001; 17(1):19–24. PMID: <u>11137736</u>
- Anderson NE, Mubanga J, Fevre EM, Picozzi K, Eisler MC, Thomas R, et al. Characterisation of the wildlife reservoir community for human and animal trypanosomiasis in the Luangwa Valley, Zambia. Ndung'u JM, editor. PLoS Negl Trop Dis. Public Library of Science; 2011 Jun; 5(6):e1211. https://doi. org/10.1371/journal.pntd.0001211 PMID: 21713019
- Auty H, Anderson NE, Picozzi K, Lembo T, Mubanga J, Hoare R, et al. Trypanosome diversity in wildlife species from the serengeti and Luangwa Valley ecosystems. PLoS Negl Trop Dis. 2012; 6(10):e1828. https://doi.org/10.1371/journal.pntd.0001828 PMID: 23094115
- Welburn SC, Picozzi K, Fèvre E, Coleman P, Odiit M, Carrington M, et al. Identification of human-infective trypanosomes in animal reservoir of sleeping sickness in Uganda by means of serum-resistanceassociated (SRA) gene. Lancet. 2001 Dec; 358(9298):2017–9. PMID: <u>11755607</u>
- Fèvre EM, Coleman PG, Odiit M, Magona JW, Welburn SC, Woolhouse ME. The origins of a new Trypanosoma brucei rhodesiense sleeping sickness outbreak in eastern Uganda. Lancet (London, England). 2001 Aug 25; 358(9282):625–8.
- Hide G. History of sleeping sickness in East Africa. Clin Microbiol Rev. 1999 Jan; 12(1):112–25. PMID: 9880477
- Legros D, Ollivier G, Gastellu-Etchegorry M, Paquet C, Burri C, Jannin J, et al. Treatment of human African trypanosomiasis—present situation and needs for research and development. Lancet Infect Dis. 2002 Jul; 2(7):437–40. PMID: 12127356

- Pépin J, Milord F. The treatment of human African trypanosomiasis. Adv Parasitol. 1994 Jan; 33:1–47. PMID: 8122565
- Trouiller P, Olliaro P, Torreele E, Orbinski J, Laing R, Ford N. Drug development for neglected diseases: a deficient market and a public-health policy failure. Lancet. 2002 Jun 22; 359(9324):2188–94. <u>https://doi.org/10.1016/S0140-6736(02)09096-7 PMID: 12090998</u>
- Barrett MP. The fall and rise of sleeping sickness. Lancet. 1999 Apr 3; 353(9159):1113–4. <u>https://doi.org/10.1016/S0140-6736(98)00416-4</u> PMID: 10209970
- Simarro PP, Franco J, Diarro A, Ruiz-Postigo JA, Jannin J. Update on field use of the available drugs for the chemotherapy of human African trypanosomiasis. Parasitology. 2012 Jun 6; 139(7):842–6. https://doi.org/10.1017/S0031182012000169 PMID: 22309684
- Welburn SC, Coleman PG, Maudlin I, Fèvre EM, Odiit M, Eisler MC. Crisis, what crisis? Control of Rhodesian sleeping sickness. Trends Parasitol. 2006 Mar; 22(3):123–8. https://doi.org/10.1016/j.pt.2006. 01.011 PMID: 16458071
- Rock KS, Torr SJ, Lumbala C, Keeling MJ, Camara O, Courtin F. Predicting the Impact of Intervention Strategies for Sleeping Sickness in Two High-Endemicity Health Zones of the Democratic Republic of Congo. Ndung?u JM, editor. PLoS Negl Trop Dis. CABI; 2017 Jan 5; 11(1):e0005162. https://doi.org/ 10.1371/journal.pntd.0005162 PMID: 28056016
- Vickerman K. Antigenic variation in trypanosomes. Nature. 1978 Jun 22; 273(5664):613–7. PMID: 661969
- Vreysen MJ, Saleh KM, Ali MY, Abdulla AM, Zhu ZR, Juma KG, et al. Glossina austeni (Diptera: Glossinidae) eradicated on the island of Unguja, Zanzibar, using the sterile insect technique. J Econ Entomol. 2000 Feb; 93(1):123–35. PMID: 14658522
- Thompson JW, Mitchell M, Rees RB, Shereni W, Schoenfeld AH, Wilson A. Studies on the efficacy of deltamethrin applied to cattle for the control of tsetse flies (Glossina spp.) in southern Africa. Trop Anim Health Prod. 1991 Dec; 23(4):221–6. PMID: 1796525
- Löhr KF, Omukuba JN, Njogu AR, Maloo SH, Gisemba F, Okedi T, et al. Investigation of the efficacy of flumethrin pour-on for the control of high tsetse and trypanosomiasis challenge in Kenya. Trop Med Parasitol. 1991 Jun; 42(2):131–4. PMID: 1896770
- Adam Y, Cecchi G, Kgori PM, Marcotty T, Mahama CI, Abavana M, et al. The sequential aerosol technique: a major component in an integrated strategy of intervention against Riverine Tsetse in Ghana. PLoS Negl Trop Dis. Public Library of Science; 2013 Jan 14; 7(3):e2135. https://doi.org/10.1371/ journal.pntd.0002135 PMID: 23516662
- Vickerman K, Tetley L, Hendry KA, Turner CM. Biology of African trypanosomes in the tsetse fly. Biol cell. 1988; 64(2):109–19. PMID: 3067793
- Vickerman K. Developmental Cycles and Biology Pathogenic Trypanosomes. Br Med Bull. Oxford University Press; 1985 Jan 1; 41(2):105–14. PMID: 3928017
- Simarro PP, Diarra A, Ruiz Postigo JA, Franco JR, Jannin JG. The human African trypanosomiasis control and surveillance programme of the World Health Organization 2000–2009: the way forward. PLoS Negl Trop Dis. Public Library of Science; 2011 Jan 22; 5(2):e1007. https://doi.org/10.1371/journal.pntd. 0001007 PMID: 21364972
- Anderson NE, Mubanga J, Machila N, Atkinson PM, Dzingirai V, Welburn SC. Sleeping sickness and its relationship with development and biodiversity conservation in the Luangwa Valley, Zambia. Parasit Vectors. 2015 Dec 15; 8(1):224.
- 23. Armstrong AJ, Blackmore A. Tsetse flies should remain in protected areas in KwaZulu-Natal. Koedoe. AOSIS Publishing; 2017 Jan 30; 59(1):1–12.
- Cecchi G, Courtin F, Paone M, Diarra A, Franco JR, Mattioli RC, et al. Mapping sleeping sickness in Western Africa in a context of demographic transition and climate change. Parasite. 2009 Jun 15; 16 (2):99–106. https://doi.org/10.1051/parasite/2009162099 PMID: 19585887
- Courtin F, Jamonneau V, Duvallet G, Garcia A, Coulibaly B, Doumenge JP, et al. Sleeping sickness in West Africa (1906–2006): changes in spatial repartition and lessons from the past. Trop Med Int Heal. 2008 Apr 6; 13(3):334–44.
- Alderton S, Noble J, Atkinson P. Simulating Sleeping Sickness: a two host agent-based model. In: Advances in Artificial Life, ECAL 2013. MIT Press; 2013. p. 27–34.
- 27. Alderton S, Macleod ET, Anderson NE, Schaten K, Kuleszo J, Simuunza M, et al. A Multi-Host Agent-Based Model for a Zoonotic, Vector-Borne Disease. A Case Study on Trypanosomiasis in Eastern Province, Zambia. Azman AS, editor. PLoS Negl Trop Dis. Unwin; 2016 Dec 27; 10(12):e0005252. https:// doi.org/10.1371/journal.pntd.0005252 PMID: 28027323
- Muller G, Grébaut P, Gouteux J-P. An agent-based model of sleeping sickness: simulation trials of a forest focus in southern Cameroon. C R Biol. 2004; 327(1):1–11. PMID: 15015750

- Vale GA, Torr SJ. User-friendly models of the costs and efficacy of tsetse control: application to sterilizing and insecticidal techniques. Med Vet Entomol. 2005 Sep; 19(3):293–305. https://doi.org/10.1111/j. 1365-2915.2005.00573.x PMID: 16134978
- Madsen T, Wallace DI, Zupan N. Seasonal Fluctuations in Tsetse Fly Populations and Human african Trypanosomiasis: A Mathematical Model. In: Mondaini RP, editor. BilOMAT 2012: International Symposium on Mathematical and Computational Biology. Singapore: World Scientific Publishing Company; 2013. p. 56–69.
- Grébaut P, Girardin K, Fédérico V, Bousquet F. Simulating the elimination of sleeping sickness with an agent-based model. Parasite. EDP Sciences; 2016; 23:63. <u>https://doi.org/10.1051/parasite/2016066</u> PMID: 28008825
- Lin S, DeVisser MH, Messina JP. An agent-based model to simulate tsetse fly distribution and control techniques: a case study in Nguruman, Kenya. Ecol Modell. 2015 Oct 24; 314:80–9. https://doi.org/10. 1016/j.ecolmodel.2015.07.015 PMID: 26309347
- Barclay HJ, Vreysen MJB. A dynamic population model for tsetse (Diptera: Glossinidae) area-wide integrated pest management. Popul Ecol. Springer Japan; 2011 Jan 9; 53(1):89–110.
- Farmer JD, Foley D. The economy needs agent-based modelling. Nature. 2009; 460(7256):685–6. https://doi.org/10.1038/460685a PMID: 19661896
- **35.** Axelrod R. The complexity of cooperation: Agent-based models of competition and collaboration. Princeton University Press; 1997.
- Epstein JM. Agent-based computational models and generative social science. Complexity. John Wiley & Sons, Inc.; 1999 May; 4(5):41–60.
- Phelps RJ, Burrows PM. Puparial Duration in Glossina Morsitans Orientalis Under Conditions of Constant Temperature. Entomol Exp Appl. Blackwell Publishing Ltd; 1969 Mar; 12(1):33–43.
- Pollock JN. Training Manual for Tsetse Control Personnel. Vol. 2: Ecology and Behaviour of Tsetse. Rome: Food and agricultural Organization of the United Nations; 1982.
- Phelps RJ, Burrows PM. Lethal Temperatures for Puparia of Glossina Morsitans Orientalis. Entomol Exp Appl. Blackwell Publishing Ltd; 1969 Mar; 12(1):23–32.
- Hargrove JW. Reproductive rates of tsetse flies in the field in Zimbabwe. Physiol Entomol. Blackwell Publishing Ltd; 1994 Dec; 19(4):307–18.
- Hargrove JW. Tsetse population dynamics. In: Maudlin I, Holmes P, Miles M, editors. The Trypanosomiases. Wallingford: CABI; 2004. p. 113–37.
- 42. Mitchell M. Complex systems: Network thinking. Artif Intell. 2006 Dec; 170(18):1194–212.
- 43. Bar-Yam Y. Dynamics of complex systems. Addison-Wesley; 1997.
- 44. Roche B, Guégan J-F, Bousquet F. Multi-agent systems in epidemiology: a first step for computational biology in the study of vector-borne disease transmission. BMC Bioinformatics. 2008 Jan; 9(1):435.
- 45. Bonabeau E. Agent-based modeling: methods and techniques for simulating human systems. Proc Natl Acad Sci U S A. 2002 May 14; 99 Suppl 3:7280–7.
- 46. Stroud P, Valle S Del, Sydoriak S, Riese J, Mniszewski S. Spatial Dynamics of Pandemic Influenza in a Massive Artificial Society. J Artif Soc Soc Simul. Journal of Artificial Societies and Social Simulation; 2007 Oct 31; 10(4).
- Yang Y, Atkinson P, Ettema D. Individual space-time activity-based modelling of infectious disease transmission within a city. J R Soc Interface. 2008 Jul 6; 5(24):759–72. <u>https://doi.org/10.1098/rsif.</u> 2007.1218 PMID: 17999949
- Auchincloss AH, Diez Roux A V. A new tool for epidemiology: the usefulness of dynamic-agent models in understanding place effects on health. Am J Epidemiol. 2008 Jul 1; 168(1):1–8. <u>https://doi.org/10. 1093/aje/kwn118</u> PMID: 18480064
- DDDAC. Research Update: Situation Analysis Zambia. (accessed January 2016) See http://steps-centre.org/wp-content/uploads/Zambia_RU_230913_WEB.pdf. 2013;
- 50. Yang J, Prince SD. Remote sensing of savanna vegetation changes in Eastern Zambia 1972–1989. Int J Remote Sens. Taylor & Francis Group; 2000 Jan; 21(2):301–22.
- Climate-data.org. Climate: Mfuwe Airport https://en.climate-data.org/location/1051466/ [Accessed 15/ 8/17] [Internet]. 2017 [cited 2017 Jun 20]. Available from: https://en.climate-data.org/location/1051466/
- Astle WL, Webster R, Lawrance CJ. Land Classification for Management Planning in the Luangwa Valley of Zambia. J Appl Ecol. 1969 Aug; 6(2):143.
- 53. Stephens JWW, Fantham HB. On the Peculiar Morphology of a Trypanosome from a Case of Sleeping Sickness and the Possibility of Its Being a New Species (T. rhodesiense). Proc R Soc B Biol Sci. 1910 Nov 26; 83(561):28–33.

- Kinghorn A, Yorke W, Lloyd L. Final Report of the Luangwa Sleeping Sickness Commission of the British South Africa Company 1911–1912. Ann Trop Med Parasit. 1913 Jun 10; 7(2):183–302.
- 55. Buyst H. The epidemiology, clinical features, treatment and history of sleeping sickness on the northern edge of the Luangwa fly belt. Med J Zambia. 1974; 4:2–12.
- 56. Buyst H. The epidemiology of sleeping sickness in the historical Luangwa valley. Ann Soc Belg Med Trop (1920). 1977 Jan; 57(4–5):349–59.
- 57. IPCC. Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. In: Barros VR, Field CB, Dokken DJ, Mastrandrea MD, Mach KJ, Bilir TE, et al., editors. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA,: Cambridge University Press; 2014. p. 688.
- Gannon C, Kandy D, Turner J, Kumar I, Pilli-Sihvola K, Sisala Chanda F. Project Report: Future Climate for Africa. Near-term climate change in Zambia. The Hague: Red Cross/Red Crescent Climate Centre; 2014.
- Alderton S, Noble J, Schaten K, Welburn SC, Atkinson PM. Exploiting Human Resource Requirements to Infer Human Movement Patterns for Use in Modelling Disease Transmission Systems: An Example from Eastern Province, Zambia. Amblard F, editor. PLoS One. Public Library of Science; 2015 Sep 30; 10(9):e0139505. https://doi.org/10.1371/journal.pone.0139505 PMID: 26421926
- **60.** Leak SGA. Tsetse biology and ecology: their role in the epidemiology and control of trypanosomosis. Oxford and New York: CABI Publishing; 1998.
- Ackley SF, Hargrove JW. A dynamic model for estimating adult female mortality from ovarian dissection data for the tsetse fly Glossina pallidipes Austen sampled in Zimbabwe. Valenzuela JG, editor. PLoS Negl Trop Dis. Public Library of Science; 2017 Aug 30; 11(8):e0005813. <u>https://doi.org/10.1371/journal.pntd.0005813</u> PMID: 28854189
- Jackson CHN. Some New Methods in the Study of Glossina morsitans. Proc Zool Soc London. 1937 Aug 21; 106(4):811–96.
- Rogers DJ, Randolph SE. Estimation of rates of predation on tsetse. Med Vet Entomol. 1990 Apr; 4 (2):195–204. PMID: 2132983
- Linden MEL. Effect of temperature on the reproduction of Glossina pallidipes, with reference to G. m. morsitans. Entomol Exp Appl. Blackwell Publishing Ltd; 1984 Jul 1; 36(1):77–83.
- Hargrove JW, Ouifki R, Ameh JE. A general model for mortality in adult tsetse (Glossina spp.). Med Vet Entomol. 2011 Dec; 25(4):385–94. https://doi.org/10.1111/j.1365-2915.2011.00953.x PMID: 21414021
- Reliasoft Corporation. Failure Rate of a Series System Using Weibull++. Reliability Hotwire. 2010; (108):http://www.weibull.com/hotwire/issue108/relbasics1.
- Hargrove JW. Age-dependent changes in the probabilities of survival and capture of the tsetse, Glossina morsitans morsitans Westwood. Int J Trop Insect Sci. Cambridge University Press; 1990 Sep 19; 11(3):323–30.
- Van Den Bossche P, De Deken R. Seasonal variations in the distribution and abundance of the tsetse fly, Glossina morsitans morsitans in eastern Zambia. Med Vet Entomol. Blackwell Science Ltd; 2002 Jun; 16(2):170–6. PMID: 12109711
- Vale GA, Hargrove JW. Predicting the effect of climate change on the abundance and distribution of tsetse flies. Q Updat Epidemiol from South African Cent Epidemiol Model Anal (SACEMA). 2015;
- 70. Mwiinde AM, Simuunza M, Namangala B, Chama-Chiliba CM, Machila N, Anderson N, et al. Estimating the economic and social consequences for patients diagnosed with human African trypanosomiasis in Muchinga, Lusaka and Eastern Provinces of Zambia (2004–2014). Infect Dis Poverty. 2017 Dec 10; 6 (1):150. https://doi.org/10.1186/s40249-017-0363-6 PMID: 29017597
- Odiit M, Coleman PG, Liu W-C, McDermott JJ, Fèvre EM, Welburn SC, et al. Quantifying the level of under-detection of Trypanosoma brucei rhodesiense sleeping sickness cases. Trop Med Int Health. 2005 Sep; 10(9):840–9. https://doi.org/10.1111/j.1365-3156.2005.01470.x PMID: 16135190
- **72.** Torr SJ, Hargrove JW. Behaviour of tsetse (Diptera: Glossinidae) during the hot season in Zimbabwe: the interaction of micro-climate and reproductive status. Bull Entomol Res. 1999 Apr 9; 89(4).